Primary care management of cardiovascular risk for people with severe mental illnesses: the Primrose research programme including cluster RCT

David Osborn,^{1,2}* Alexandra Burton,¹ Kate Walters,³ Lou Atkins,⁴ Thomas Barnes,⁵ Ruth Blackburn,¹ Thomas Craig,⁶ Hazel Gilbert,³ Ben Gray,⁷ Sarah Hardoon,³ Samira Heinkel,¹ Richard Holt,⁸ Rachael Hunter,³ Claire Johnston,⁹ Michael King,^{1,2} Judy Leibowitz,² Louise Marston,³ Susan Michie,^{2,4} Richard Morris,³ Steve Morris,¹⁰ Irwin Nazareth,³ Rumana Omar,¹¹ Irene Petersen,³ Robert Peveler,⁸ Vanessa Pinfold,⁷ Fiona Stevenson³ and Ella Zomer³

- ¹Division of Psychiatry, Faculty of Brain Sciences, University College London, London, UK
- ²Camden and Islington NHS Foundation Trust, St Pancras Hospital, London, UK
 ³Department of Primary Care and Population Health, University College London, London, UK
- ⁴Centre for Behaviour Change, Department of Clinical, Educational and Health Psychology, Division of Psychology and Language Sciences, Faculty of Brain Sciences, University College London, London, UK
- ⁵Faculty of Medicine, Department of Medicine, Imperial College London, London, UK
- ⁶Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK
- ⁷The McPin Foundation, London, UK
- ⁸Human Development and Health Academic Unit, Faculty of Medicine, University of Southampton, Southampton, UK
- ⁹School of Health and Education, Faculty of Professional and Social Sciences, Middlesex University, London, UK
- ¹⁰Department of Allied Health Research, University College London, London, UK
- ¹¹Department of Statistical Science, University College London, London, UK

*Corresponding author d.osborn@ucl.ac.uk

Declared competing interests of authors: Thomas Craig reports personal fees from Otsuka Pharmaceuticals Europe Ltd, Wexham, UK. Richard Holt reports personal fees from Eli Lilly and Company (IN, USA), Jannssen Pharmaceutica (Beerse, Belgium), Sunovion, Pharmaceuticals Inc. (MA, USA) and Otsuka Pharmaceuticals Europe Ltd and is a member of the Health Technology Assessment (HTA) Women and Children's Health Panel. Thomas Barnes reports personal fees from Sunovion Pharmaceuticals Inc., Newron Pharmaceuticals SpA (Bresso, Italy), and Otsuka Pharmaceutical Co., Ltd (Tokyo, Japan) and Lundbeck, Copenhagen, Denmark Kate Walters was a member of the Disease Prevention Panel and the Primary Care Commissioning Panel. Susan Michie was a member of the HTA Pandemic Influenza Board. Michael King was a member of the National Institute for Health Research (NIHR)-funded Clinical Trials Units (CTUs), Rapid Trials and Add-on Studies Board. Irwin Nazareth's membership includes CTUs funded by NIHR, the Disease Prevention Panel, HTA Commissioning Board, HTA Commissioning Sub-board (Expression of Interest) and the HTA Primary Care Themed Call. Rumana Omar was a member of the HTA General Board. Steve Morris sat as a member on the Health Services and Delivery Research (HSDR) board, HSDR Commissioning Board, HSDR Evidence Synthesis Sub-board, HTA Commissioning Board and Public Health Research Research Funding Board.

Published April 2019 DOI: 10.3310/pgfar07020

Plain English summary

Primrose research programme including cluster RCT

Programme Grants for Applied Research 2019; Vol. 7: No. 2 DOI: 10.3310/pgfar07020

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

People with severe mental illnesses (SMI), including schizophrenia and bipolar disorder, die earlier from heart attacks and strokes. This programme of work explored how to best prevent heart attacks or stroke in people with SMI.

First, a new computerised tool for general practitioner (GP) practices was developed to predict heart attacks for people with SMI. This was then compared with existing computerised tools and it was found that this new tool was better. Second, staff and patients were asked for their views on what a service for reducing the risk of heart attacks should look like. Existing research was reviewed and national data were used to look at statin prescribing (a medication for lowering levels of cholesterol) for people with SMI. It was found that statins decreased levels of cholesterol, that physical activity and diet lowered weight, and that accessing Stop Smoking Services and medications reduced smoking.

This evidence was used to develop a new service to lower levels of cholesterol in people with SMI. Patient experts were involved in decisions of what to include in the service and in training nurses to review medication, diet, physical activity, smoking, blood pressure or diabetes mellitus in 8–12 appointments at GP practices.

Third, whether or not patients had lower levels of cholesterol after receiving the new service at their GP practice when compared with patients receiving GP usual care was assessed. The first large trial was delivered to improve physical health in SMI in general practices. A total of 327 patients took part in 76 general practices: 38 practices had the new service and 38 did not. The service was delivered well and people attended. Levels of cholesterol went down for both those who did and those who did not receive the new service. There was no difference between the groups in terms of levels of cholesterol or other physical measures. The care provided by the general practices that did not receive the new Primrose service may have been better than normal general practice care. These general practices still had to identify people with raised CVD risk factors to take part in the study who might not have been identified if the general practice had not taken part in the study. This may explain why no differences were found. People who received the new service had fewer inpatient mental health admissions over the year.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Osborn *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was set up in 2006 to produce independent research findings that will have practical application for the benefit of patients and the NHS in the relatively near future. The Programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director.

The programme is a national response mode funding scheme that aims to provide evidence to improve health outcomes in England through promotion of health, prevention of ill health, and optimal disease management (including safety and quality), with particular emphasis on conditions causing significant disease burden.

For more information about the PGfAR programme please visit the website: http://www.nihr.ac.uk/funding/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0609-10156. The contractual start date was in March 2011. The final report began editorial review in September 2017 and was accepted for publication in August 2018. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, the PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PHR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Osborn *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk